Check for updates

Citation: Pekelharing JE, Gatluak F, Harrison T, Maldonado F, Siddiqui MR, Ritmeijer K (2020) Outcomes of visceral leishmaniasis in pregnancy: A retrospective cohort study from South Sudan. PLoS Negl Trop Dis 14(1): e0007992. https://doi. org/10.1371/journal.pntd.0007992

Editor: Carlos Franco-Paredes, Hospital Infantil de Mexico Federico Gomez, UNITED STATES

Received: August 20, 2019

Accepted: December 12, 2019

Published: January 24, 2020

Copyright: © 2020 Pekelharing et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: MSF has a managed access system for data sharing that respects MSF's legal and ethical obligations to its patients to collect, manage and protect their data responsibility. Ethical risks include, but are not limited to, the nature of MSF operations and target populations being such that data collected are often highly sensitive. Data are available on request in accordance with MSF's data sharing policy (available at: http://fieldresearch.msf.org/msf/ handle/10144/306501). Requests for access to data should be made to data.sharing@msf.org. RESEARCH ARTICLE

Outcomes of visceral leishmaniasis in pregnancy: A retrospective cohort study from South Sudan

Judith E. Pekelharing^{1*}, Francis Gatluak², Tim Harrison¹, Fernando Maldonado³, M. Ruby Siddiqui⁴, Koert Ritmeijer¹

1 Médecins Sans Frontières, Amsterdam, Netherlands, 2 Médecins Sans Frontières, Lankien, South Sudan, 3 Royal Tropical Institute, Amsterdam, Netherlands, 4 Médecins Sans Frontières, London, United Kingdom

* jepekelharing@gmail.com

Abstract

Introduction

Visceral leishmaniasis (VL) is endemic in South Sudan, where outbreaks occur frequently. Because of changes in the immune system during pregnancy, pregnant women are considered particularly vulnerable for developing complications of VL disease, including opportunistic infections. There is limited evidence available about clinical aspects and treatment outcomes of VL in pregnancy. We describe characteristics, maternal and pregnancy outcomes from a cohort of pregnant women with VL.

Methods

We conducted a retrospective analysis using routine programme data from a MSF health facility in Lankien, Jonglei State, South Sudan, between Oct 2014 and April 2018. Records were extracted of women diagnosed with VL while pregnant, and those symptomatic during pregnancy but diagnosed during the first two weeks postpartum. Records were matched with a random sample of non-pregnant women of reproductive age (15–45 years) with VL from the same period.

Results

We included 113 women with VL in pregnancy, and 223 non-pregnant women with VL. Women with VL in pregnancy presented with more severe anaemia, were more likely to need blood transfusion (OR 9.3; 95%CI 2.5–34.2) and were more often prescribed antibiotics (OR 6.0; 95%CI 3.4–10.6), as compared to non-pregnant women with VL. Adverse pregnancy outcomes, including miscarriage and premature delivery, were reported in 20% (16/ 81) where VL was diagnosed in pregnancy, and 50% 13/26) where VL was diagnosed postpartum. Postpartum haemorrhage was common. Pregnant women were more likely to require extension of treatment to achieve cure (OR 10.0; 95%CI 4.8–20.9), as compared to non-pregnant women with VL. Nevertheless, overall initial cure rates were high (96.5%) and mortality was low (1.8%) in this cohort of pregnant women with VL. **Funding:** The author(s) received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

Conclusion

This is the largest cohort in the literature of VL in pregnancy. Our data suggest that good maternal survival rates are possible in resource-limited settings, despite the high incidence of complications.

Author summary

Visceral leishmaniases (VL), also known as 'Kala azar', is a parasitic disease transmitted by sand flies. South Sudan is one of the countries in the world with the highest disease burden of VL. Existing evidence about treatment and outcomes of VL in pregnancy is limited. We described outcomes of pregnant women with VL and women diagnosed with VL in the first two weeks postpartum. Pregnant women with VL were treated with liposomal amphotericin B (AmBisome) in this cohort. Survival rates and initial cure rates of women with VL in pregnancy treated in this resource constrained setting were high. However, these patients often required extension of treatment to achieve cure after the initial standard dose of 30mg/kg AmBisome in six doses. Careful evaluation of patients is needed at the end of the standard treatment regimen, in order to ensure proper (parasitological) cure before discharge. We reported a high incidence of complications (including severe anaemia with a need for blood transfusion), adverse pregnancy outcomes (including miscarriage and premature delivery), and obstetric complications. This research adds to the existing body of knowledge concerning VL in pregnancy, although several questions about this condition remain unanswered. Topics for future research include the immunological response to leishmania infection in pregnancy, and maternal, pregnancy and neonatal outcomes after discharge from VL treatment.

Introduction

Visceral Leishmaniasis (VL), also known as Kala Azar, is endemic in South Sudan, where it is caused by the protozoan *Leishmania donovani* parasite, and transmitted by phlebotomine sand-flies. [1,2].

South Sudan is among the countries with the highest VL burden worldwide, and outbreaks occur regularly, often fueled by civil unrest and mass displacement [3,4].

VL affects the lymphatic and reticuloendothelial system, and patients typically present with fever, malaise, weight loss and anorexia. Clinical signs are splenomegaly, hepatomegaly, wasting and lymphadenopathy, and the disease causes anaemia and sometimes pancytopenia. Complications occur from anaemia, bleeding, malnutrition and intercurrent infections. Without treatment, the disease can progress rapidly in weeks or months and is typically fatal[2,5].

The current first-line treatment in East Africa for uncomplicated VL outside pregnancy is a 17-day course with a combination therapy of intramuscular injections of a pentavalent antimonial sodiumstibogluconate (SSG) and aminoglycoside paromomycin (PM)[2]. Patients with a high risk of (fatal) toxicity or intolerability to pentavalent antimonials are treated with intravenous infusions of liposomal amphotericin B (AmBisome), as this is much better tolerated. Indications for AmBisome treatment include pregnancy, extreme ages, HIV-coinfection and severe VL. Initial cure rates of 93–97% [2,6] and mortality of 2.8%-3% have been reported in large patient cohorts treated in South Sudan between 2009 and 2015 (with SSG/PM as first-line treatment and AmBisome for specific indications)[7,8]. Pregnant women are a particular patient group, considering the modulated maternal immunological system during pregnancy. Pregnancy is described as a pro-inflammatory and anti-inflammatory condition, depending upon the stage of gestation[9]. Current insights on how the immune system interacts with leishmania infection in pregnancy are based on animal studies. T-cell balances and different resulting cytokine patterns are suggested to explain different scenarios of the maternal response to leishmania infection. For example the Th1-predominant inflammatory immune response is associated with both rejection of pregnancy and control of intracellular Leishmania infections, while a Th2-predominant response is associated with preservation of pregnancy and poor Leishmania infection control[10]. No studies are available looking at the particular susceptibility for pregnant women to develop clinical VL disease after infection.

Treatment of VL in pregnancy is indicated, because the threat of a fatal outcome for the mother, the fetus or the newborn due to untreated VL is much greater than the risk related to adverse drug effects[2]. Several clinical studies and case series on VL in pregnancy are available, but all with relatively small patient numbers. Vertical transmission and adverse pregnancy outcomes have been reported, but the prevalence of congenital VL after maternal disease during pregnancy is not known[11,12]. High rates of miscarriages are reported in pregnant women treatment with SSG, up to 57–69% in some studies[13,14]. Survival rates for pregnant women treated for VL vary; some studies report cure rates as high as 100%[13,14] but case fatality rates of 18% are reported in other settings[15]. AmBisome is now considered the safest therapeutic option for VL in pregnancy and it is the recommended treatment for this group in South Sudan[2,16].

The aim of this study is to describe maternal and perinatal outcomes of patients with VL in pregnancy.

Methods

A retrospective analysis was conducted, using routinely collected data of a VL cohort from a treatment facility in Lankien, Jonglei state, in the north eastern part of South Sudan. The hospital facility has been run by Médecins Sans Frontières (MSF) since 1993, and provides primary outpatient and inpatient health care services, comprehensive maternity care, and treatment for TB, HIV, and tropical diseases including VL. The population in the catchment area is a pastoralist population who practice cattle husbandry and agriculture. The facility is the only hospital in a large region, and located in an area with high VL endemicity, however, access to the health services is often difficult due to poor road conditions, flooding during the rainy season and insecurity. Between 1,000 and 4,600 VL patients per year have been treated in this facility over the past 10 years. In 2017 almost 50% of all reported VL cases in South Sudan were treated in Lankien.

Diagnosis and treatment

The following diagnostic procedure was pursued for all patients regardless of their pregnancy status. Patients meeting the clinical case definition (history of fever more than 2 weeks and splenomegaly and/or lymphadenopathy and/or wasting (BMI <16 kg/m² for ages >18 years or weight-for-height <-2 Z score for ages \leq 18 years)) underwent further laboratory evaluation to confirm VL following a diagnostic algorithm. Diagnostic tests included the rK39 rapid diagnostic test (IT-Leish, Bio-Rad laboratories, USA) for patients with no prior VL treatment history (suspect primary VL), with a positive result confirming VL. Those testing negative were screened with the leishmania direct agglutination test (DAT, Royal Tropical Institute, Amsterdam, The Netherlands) and a high titer (\geq 1:6,400) confirmed VL. Those with an intermediate

DAT titer (1:800–1:3,200) underwent tissue aspiration (spleen or lymph node) for microscopy. Demonstration of leishmania amastigotes confirmed VL. Tissue aspirate microscopy was used as the initial diagnostic test for all patients with a prior history of VL treatment (suspected VL relapse). A clinical diagnosis of relapse VL was made in patients with a contra-indication for spleen aspiration and negative lymph node results, if there was persistent clinical suspicion for VL and no likely alternative diagnosis. All confirmed VL patients were routinely tested for HIV and malaria with rapid diagnostic tests. The protocol indicated a urine pregnancy test for confirmed female VL patients of reproductive age before start of treatment. Patients were evaluated for possible tuberculosis co-infection based on clinical signs and symptoms and the response to VL treatment. Tuberculosis co-infection was confirmed by sputum microscopy or GeneXpert (cartridge-based nucleic acid amplification test) analysis.

Non-pregnant women with uncomplicated VL were treated with the combination of SSG (20 mg/kg) and PM (15 mg/kg) given on an ambulatory basis over 17 days with daily intramuscular injections. All pregnant women diagnosed with VL and non-pregnant patients with severe VL (according to the MSF treatment protocol) were treated with AmBisome by 6 IV infusions of 5 mg/kg on alternate days.

All patients received nutritional support in the form of therapeutic food, and treatment for dehydration and secondary infections if indicated. The indication for treatment with (intravenous) antibiotics was based on clinical suspicion of a bacterial infection requiring treatment with antibiotics, as diagnostic facilities in the treatment setting were limited. Blood transfusion was considered in case of haemoglobin (Hb) <5.0g/dL, or <7.0g/dL if there were signs of decompensation.

Study design and data source

Women who were pregnant at time of VL diagnosis were identified from the routine electronic database of VL patients (group 1). A second study group included patients who were already symptomatic during pregnancy but diagnosed with VL within the first two weeks after spontaneous abortion or delivery (either home delivery or hospital delivery). This group therefore includes patients with VL in pregnancy, but diagnosed postpartum (group 2). These patients were identified by screening maternity discharge files. Finally, from the VL database, a random selection of non-pregnant VL-positive women of reproductive age (between 15 and 45 years) were selected to serve as comparison group (group 3) with a ratio of 1:2 for pregnant: non-pregnant.

Patient medical records were reviewed to verify data from the database and to complete missing data. Patient characteristics, complications during treatment and treatment outcomes were compared between the different groups.

Inclusion and exclusion criteria

Patients discharged from VL treatment between October 2014 and July 2016 and in the period March-April 2018 were considered for analysis. No medical records were available for the period September 2016 until February 2018, as these files had been accidentally destroyed. Records with no information on pregnancy status were excluded. Women of reproductive age, with either a negative urine pregnancy test or a note on the admission card indicating that the patient was not pregnant, were considered to be not pregnant.

Data collection

Data was transferred into a purpose-made database created using Epidata software (version 4.4.2.0). Key dates, demographic, anthropometric, diagnostic and clinical characteristics of patients, treatment regime, complications during treatment and outcome were recorded.

Outcome

All patients were evaluated for cure (either by clinical improvement or parasitological cure) after completing the standard treatment regimen. 'Clinical cure' was by judgement of the clinician, based on the following signs of cure: absence of fever, regression of spleen, weight gain, increase in Hb and improved appetite. Parasitological cure means that a test-of-cure (ToC) by aspirate microscopy was performed at day 21–28 after initiating treatment to confirm cure at the end of treatment. According to the protocol, a ToC was indicated for patients with an increased risk of treatment failure or relapse (i.e. patients with a prior episode of VL, or patients with inadequate or doubtful clinical response, or patients with a known immune-suppressive condition (HIV or TB co-infection), and pregnant patients up to the first 6 months postpartum).

Treatment was extended if no initial cure was achieved after the standard dose treatment. This is presented as 'need for extended treatment'. The decision to extend treatment could either be based on a positive ToC, or no/inadequate clinical response. If the treatment was extended based on a positive ToC, the ToC was repeated every week until negative. If treatment was extended based on clinical assessment, a ToC was done to assess the effect of extended treatment.

Treatment outcomes were defined as 'initial cure at discharge', 'defaulter' or 'death'. There was no systematic follow up after discharge, because patients returned to areas remote from the treatment centre, and therefore we have limited data about definite cure at six months post-treatment. A defaulter was defined as a patient who did not complete the prescribed treatment, left against medical advice, and had an unknown outcome.

The following pregnancy and neonatal outcomes were recorded: uncomplicated pregnancy outcome (defined as a live term delivery during treatment, or when the patient was still pregnant at time of discharge); adverse pregnancy outcome (defined as first trimester abortion, immature or premature delivery <37 weeks' gestation, or stillbirth); birth weight. Postpartum haemorrhage and surgical evacuation for retained products of conception were recorded as obstetric/postpartum complications.

For each pregnant women gestation of pregnancy was recorded using three categories: first trimester (<13 weeks' gestation), second trimester (13–27 weeks' gestation) and third trimester (>27 weeks' gestation). Often the exact gestation was not known, but an estimation was made based on history of last menstruation and assessment of the abdomen on admission and/or assessment of the baby postpartum.

Statistical analysis

Categorical variables are presented using absolute numbers and percentages. Continuous variables are summarised using medians (for non-normally distributed data) and ranges. Where appropriate, differences in proportions were measured using Fisher's exact test and Pearson χ^2 test and the results expressed as odds ratios with 95% confidence intervals. Differences in continuous variables were assessed using non-parametric tests (Mann Whitney U test) as the data were non-normally distributed. Data analysis was conducted using SPSS Statistics (version 25).

Ethics

This research fulfilled the exemption criteria set by the MSF Ethical Review Board (ERB) for a *posteriori* analyses of routinely collected clinical data, and thus did not require additional clearance from the MSF-ERB. It was conducted with permission from the Medical Director of MSF-Operational Centre Amsterdam. All data were anonymised before analysis.

Results

The total cohort of female VL patients in the study period aged 15–45 years was 639. From the database with routinely collected data and by screening maternity discharge files, 96 patients were identified as VL in pregnancy. Seven (7) pregnant patients were excluded because records could not be found. Two additional patients were excluded; one turned out to be non-pregnant, and another one was diagnosed more than 2 weeks post-partum, resulting in 87 pregnant patients with VL included in group 1. Out of the 283 patients that were randomly selected to form the comparison group, 223 were included in group 3 (non-pregnant). Twenty-six (26) were included in group 2 after record review confirming that they had VL diagnosed within the first two weeks postpartum. Thirty-four (34) records were excluded from analysis: 15 because their files could not be found; 12 because the pregnancy status was unclear/not recorded in the file; and 7 were excluded because after record review it turned out they were lactating women with unknown delivery date. See Fig 1 for an overview of the study flow diagram.

Patient characteristic

Patient characteristics on admission are presented in Table 1.

Of the patients with VL in pregnancy (group 1 and 2 combined), 26.8% were in the first trimester of pregnancy at the time of diagnosis, 35.1% in the second trimester and 38.1% in the third trimester. Among the patients in the first trimester, four patients had symptoms for more than one month. These patients could possibly have been symptomatic before they became pregnant. The remaining patients (99/113) were pregnant before they became symptomatic with VL. Parity of pregnant VL patients (group 1) was distributed as follows: nullipara: 17.2%, subsequent pregnancies: 82.8% (range para 1–8).

The great majority of patients with primary VL (93%) was diagnosed with a positive rk39 rapid diagnostic test. A small group of primary VL patients required further diagnostic follow up. All VL relapse patients were diagnosed using lymphnode or spleen aspirate microscopy.



https://doi.org/10.1371/journal.pntd.0007992.g001

	VL in p	regnancy	Comparison group	VL in pregnancy (group 1 and 2 combined) compared to	
	N = 113		N = 223	Comparison group	
	Group 1 Group 2		Group 3		
	Pregnant	Postpartum	Non-pregnant		
	N = 87	N = 26	N = 223		
Age in years	Median 24, IQR 20–29	range 14–40,	Median 22, range 15–45, IQR 16–31	p 0.446	
Duration of symptoms					
• Up to 4 weeks	76 (91.6%)	24 (96%)	195 (89.9%)	p 0.543	
• > 4 weeks	7 (8.4%)	1 (4.0%)	22 (10.1%)		
Missing data	4	1	6		
Splenomegaly					
• No splenomegaly	17 = 30.9%	8 (30.8%)	63 (28.3%)	p 0.994	
• 1–4 cm	22 = 40.0%	6 (23.1%)	90 (40.4%)		
• 5-7cm	11 = 20.0%	8 (30.8%)	44 (19.7%)		
• >7cm	5 = 9.1%	4 (15.4%)	26 (11.7%)		
Missing data *	32				
Nutritional status, BMI in kg/m2**					
• < 13.0	1 = 1.4%	0 = 0%	4 = 2.9%	p < 0.001	
• 13.0-14.4	1 = 1.4%	3 = 14.3%	26 = 19.1%		
• 14.5–15.9	9 = 12.9	5 = 23.8%	39 = 28.7%		
• ≥ 16.0	59 = 84.3%	13 = 61.9%	67 = 49.3%		
Hb on admission in g/dL					
• <6.0 g/d	2 = 2.3%	2 = 7.7%	4 = 1.8%	p < 0.001	
• 6.0-6.9	10 = 11.5%	4 = 15.4%	7 = 3.1%		
• 7.0-8.9	28 = 32.2%	15 = 57.7%	68 = 30.5%		
• ≥ 9.0	47 = 54.0%	5 = 19.2%	144 = 64.6%		
Malaria	2/75 (2.7%)	2/24 (7.7%)	3/136 (2.2%)	p 1.000	
Missing data	12	2	87	OR 1.2 (0.20–7.6)	
HIV	0/67 (0%)	0/21 (0%)	0/206 (0%)	p 1.000	
Missing data	20	5	17		
Primary VL	99 (8 <mark>7.6%</mark>)		196 (87.9%)	p 1.000	
Relapse VL	14 (1 <mark>2.4</mark> %)		27 (12.1%)		
Diagnosis of primary VL by positive	97.3%	81%	94.9%	p 0.785	

Table 1. Patient characteristics on admission.

* Spleen size data were missing for 32 pregnant women in the third trimester, as spleen exam was not feasible or was unreliable in advanced pregnancy

** presented for patients ≥ 19 years

https://doi.org/10.1371/journal.pntd.0007992.t001

Thirty percent (30%) of records had no information about malaria status and 12.5% had no information about HIV status.

Complications during treatment

Recorded complications during treatment are presented in Table 2.

The most commonly reported bleeding complications were nose-bleeding and obstetric related bleeding. Bleeding after an abortion and postpartum haemorrhage was also recorded as bleeding complication in this table.

	VL in pregnancy N = 113 Group 1 Group 2		<u>Comparison</u> group	VL in pregnancy (group 1 and 2 combined) compared to Comparison group		
			N = 223			
			Group 3			
	Pregnant	Postpartum	Non-pregnant			
	N = 87	<u>N = 26</u>	<u>N = 223</u>			
Bleeding	13 = 14.9%	7 = 26.9%	11 = 4.9%	p < 0.01		
Missing data	9	1	5	OR 4.5 (2.1–9.9)		
Blood transfusion	7 (9.7%)	4 (17.4%)	3 (1.4%)	p < 0.001		
Missing data	15	3	7	OR 9.3 (2.5–34.1)		
Jaundice (on admission or during treatment)	7 (9.5%)	1 (4.0%)	6 (4.2%)	p 0.265, OR 2.0 (0.67–5.9)		
Missing data	13	1	81			
Antibiotic treatment for suspected infectious complications						
- Total (iv. or oral)	49 (66.2%)	22 (88%)	40 (29.9%)	p < 0.001		
- Intravenous	34 (45.9%)	16 (64%)	10 (7.5%)	OR 6.0 (3.4–10.6)		
Missing data	13	1	89			

Table 2. Complications during VL treatment.

https://doi.org/10.1371/journal.pntd.0007992.t002

Significantly more blood transfusions were reported for patients with VL in pregnancy, compared to the comparison group. Patients who received blood transfusion had a median lowest Hb value of 5.1 g/dL, ranging from 3.5 to 7.3 g/dL.

Antibiotics were prescribed for infectious complications, including suspected sepsis in severely ill patients, respiratory tract infection, ear infection, urinary tract infection, eye infection, skin infection, abscess, and sometimes because of ongoing fever with no identified focus. Some patients diagnosed postpartum were treated with antibiotics for fever postpartum (suspected endometritis) before VL diagnosis.

Two pregnant patients were inadvertently started on SSG/PM treatment because the clinician was unaware of their pregnancy status. Both patients had an adverse pregnancy outcome (first trimester abortion and immature delivery respectively).

Change of treatment regimen from SSG/PM to AmBisome (due to SSG toxicity, jaundice or deteriorating condition) occurred in 7 (3.6%) of non-pregnant patients and none of the patients with VL postpartum.

Treatment outcomes

Treatment outcomes are presented in Table 3.

One pregnant woman with severe VL disease died after the first dose of AmBisome. One patient died after being diagnosed with VL postpartum (associated with suspected sepsis).

For four patients with VL postpartum the treatment regimen was changed to SSG/PM after a full course of AmBisome due to unsatisfactory treatment response.

For the patients in the comparison group, the treatment regimen was changed from SSG/ PM to AmBisome in 7 cases (3.6%), but only because of SSG toxicity or deteriorating condition, and not due to treatment failure.

Initial cure rates at time of discharge were not significantly different between the groups. When comparing the patients with VL in pregnancy to the patients in the non-pregnant group who were treated with AmBisome, there was no significant difference in need for treatment extension or the total dose of AmBisome. Looking at all the groups combined, patients with relapse VL had higher odds of requiring extended treatment to achieve cure (OR 2.2, 95% CI 1.05–4.50 p 0.037).

Table 3. Treatment outcomes.

	VL in pregnancy N = 113		Comparison group	VL in pregnancy (group 1 and 2 combined) compared to Comparison group	
			N = 223		
	Group 1	Group 2	Group 3		
	Pregnant	Postpartum	Non-pregnant		
	N = 87	N = 26	N = 223		
Treatment regimen					
• AmBisome	85 (97.7%)	19 (73.1%)	27 (12.1%)	p < 0.001	
• SSG/PM	2 (2.3%)	7 (26.9%)	196 (87.9%)		
Treatment extended because no cure at the end of standard treatment					
- Total	31 (35.6%)	12 (46.1%)	22 (9.8%)	p< 0.001 OR 5.7 (3.2–10.2)	
- Based on positive ToC	10 (11.4%)	8 (30.7%)	11 (4.9%)	p< 0.001 OR 4.1 (1.9-8.6)	
- Based on no clinical cure	20 (22.9%)	3 (11.5%)	9 (4.0%)	p < 0.001OR 6.0 (2.7–13.4)	
Reason not documented	1	1	2		
Total dose AmBisome*					
No. of doses					
- 6	50 (64.9%)	8 (44.4%)	12 (48.0%)	p 0.485	
- 7-8	18 (23.4%)	4 (22.3%)	10 (40.0%)		
- 9–10	7 (9.1%)	1 (5.6%)	3 (12.0%)		
- >10	2 (2.6%)	5 (27.8%)	0		
Missing data	5	0	1		
Treatment outcome					
- Initial cure at discharge	84 (96.6%)	25 (96.2%)	218 (97.8%)	p 0.739	
- Defaulter	2 (2.3%)	0	3 (1.3%)		
- Death	1 (1.1%)	1 (3.8%)	2 (0.9%)		
Among those cured:					
- Clinical cure	3 <mark>4 (4</mark> 0.5%)	7 (28.0%)	166 (76.1%)	p < 0.001	
- Parasitological cure	50 (59.5%)	18 (72.0%)	52 (23.9%)		
Discharge Hb in g/dL					
• <6.0 g/dL	1 = 1.2%	0 = 0%	1 = 0.5%	p < 0.001	
• 6.0-6.9	12 = 14.3%	3 = 12.5%	2 = 0.9%		
• 7.0-8.9	41 = 48.8%	16 = 66.7%	37 = 17.1%		
• ≥ 9.0	30 = 35.7 <mark>%</mark>	5 = 20.8%	176 = 81.5%		
missing data	3	2	7		

* For patients who were started on AmBisome as initial treatment regimen and excluding defaulters and deaths. One dose of AmBisome is 5mg/kg.

https://doi.org/10.1371/journal.pntd.0007992.t003

Obstetric and perinatal outcomes

Data about pregnancy outcomes and obstetric complications from time of VL diagnosis until discharge from VL treatment are presented in Table 4.

Stillbirth occurred in two cases: in a patient who delivered during treatment by caesarean section in the third trimester for bleeding placenta praevia, and in a patient diagnosed postpartum who delivered on the road just before admission (unknown gestation). Insufficient data were available in the patient files to trace other neonatal outcomes such as apgar score or neonatal survival rates after birth.

Pregnancy outcomes per trimester are presented in <u>Table 5</u>, for patients who were pregnant at time of VL diagnosis. Adverse pregnancy outcomes occurred more frequently in the third

Table 4. Obstetric and pregnancy outcomes.

	Group 1	Group 2
	Pregnant	Postpartum
	N = 81	N = 26
Still pregnant at time of discharge	64 (79%)	N/A
Delivered during treatment	17 (21%)	
Adverse pregnancy outcome	16 (20%)	13 (54%)
	Among those who o	delivered
Post-partum haemorrhage	6/17 (35%)	*
Surgical evacuation for retained products of conception	5/17 (29%)	6/24 (25%)

https://doi.org/10.1371/journal.pntd.0007992.t004

trimester compared to the second trimester (p 0.001, OR 14.0, 95%CI 2.8–75.6), differences between the other trimesters were not significant. Data about trimester of pregnancy were missing in 15% (13/87) cases.

Adverse pregnancy outcomes occurred more frequently among primigravidae compared with subsequent pregnancies (OR 4.33, 95% CI 1.03–18.2), although this should be interpreted carefully as data about parity were available for only 66.7% (58/87) of cases.

Twenty-nine patients delivered in the third trimester (during treatment or before starting treatment). Data on birth weights are available for the babies of only 13 patients. The median weight was 1.9kg, range 1.2–2.6kg.

Discussion

Patient characteristics

The different patient groups were similar regarding age, duration of symptoms, and proportion of primary VL/relapse VL (no significant difference between the groups). BMI was higher among pregnant women, but this is likely to be due to the pregnancy and not representing a better nutritional status.

Hb on admission was significantly lower for pregnant women compared to non-pregnant women, and lower for postpartum women compared to pregnant women. The difference can be explained by the (physiologic) anaemia in pregnancy and blood loss during delivery.

None of the women in this cohort was diagnosed with HIV infection during routine testing. Missing data are likely due to recording omissions of negative test results. Few patients in this

	D		· · ·	
Table 5.	Pregnancy	outcomes	per tr	imester.

	Uncomplicated pregnancy outcome (pregnant on discharge or live term delivery)	Adverse pregnancy outcome (miscarriage, premature delivery < 37 wks gestation, stillbirth)
Diagnosed with VL during 1 st trimester N = 22	17 (77.3%)	5 miscarriages (22.7%)
Diagnosed with VL during 2^{nd} trimester N = 33	31 (93.9%)	2 miscarriages/immature deliveries (6.1%)
Diagnosed with VL	9 pregnant on discharge (47.4%)	9 premature deliveries (47.4%) of
during 3 rd trimester: N = 19	1 live term delivery (5.3%)	which 1 stillbirth

https://doi.org/10.1371/journal.pntd.0007992.t005

cohort were diagnosed with TB co-infection. TB-incidence in South Sudan was estimated 146/ 100,000 in 2017[17]. However, because a higher TB incidence can be expected in immune-suppressed VL patients[18], TB may have been underdiagnosed, as diagnosis of (extra-pulmonary) TB is challenging.

Also, although malaria is highly endemic in South Sudan, with 159 infections per 1,000 people at risk in 2016[19], few malaria co-infections were reported. A possible explanation could be that patients were treated for malaria in the out-patient clinic before being tested for VL. This information was not available, as patients take their out-patient medical records home. Rates of malaria-VL coinfection varied between 3.8 and 60.8% in a study from Sudan, depending on the geographical setting[20].

Detailed general population data are lacking to determine whether women in certain trimesters are more susceptible to develop VL after infection and whether susceptibility is influenced by parity.

Disease related complications

Blood transfusions were needed significantly more often in patients with VL in pregnancy (OR 9.3 (2.5–34.1)). This is something to consider when planning VL treatment services.

Jaundice (either on admission or occurring during treatment) was reported in 8% of pregnant women with VL, and is a sign impaired liver function due to severe VL, and is an indication for treatment with AmBisome also for non-pregnant patients.

Antibiotics were more often prescribed in pregnant and postpartum women compared to non-pregnant patients (OR 6.0 (3.4–10.6)). There may have been over-diagnosis of suspected bacterial infections in patients with ongoing fever after a few days of treatment. On the other hand, VL is an immune-suppressive condition and secondary infections are a known complication. Empiric antibiotic therapy can be a reasonable choice for certain patients in this resource constrained setting. It is possible that physicians were more likely to indicate empiric antibiotic treatment for pregnant women and patients admitted postpartum when an infectious focus was not identified.

Treatment outcomes

Mortality during treatment in patients with VL in pregnancy (group 1 and 2 combined) was 1.8% (2/133).

This is comparable to large VL cohorts[7,8], but lower than some other reports about VL in pregnancy[15,21]. A Sudanese study reported 18% case fatality rate among 45 pregnant patients treated with SSG in 2014–2015, mainly due to hepatic failure, bleeding manifestation, and severe anaemia/heart failure[15]. In another study from Sudan where pregnant women with VL were treated with SSG, the main cause of dead was hepatic encephalopathy[21]. In our study cohort it is likely that especially the severely ill patients, including the patients with jaundice, have benefited from treatment with AmBisome, and blood-transfusion may have been a live-saving intervention in some cases.

Case fatality rate in all non-pregnant female patients in reproductive age in the cohort was 2.8% during the study period (source: database with routinely collected data), which is comparable to previous reports [7,8].

For the patients with VL in pregnancy, treatment was more often extended compared to the non-pregnant comparison group (OR 5.7 (3.2–10.2)). For some patients treated with AmBisome, the standard treatment of 30mg/kg divided in six doses was extended to up to 12 doses (60mg/kg) to achieve cure. There was no significant difference in mean total dose of AmBisome between patients with VL in pregnancy and patients treated with AmBisome in the

comparison group. The patients in the comparison group treated with AmBisome often had severe disease, which could explain why many of them also needed extended treatment. Higher frequency of extended treatment in pregnant women could partly be explained by the fact that pregnant women are routinely tested for parasitological cure, whereas for non-pregnant women this is only done in case of suspected lack of cure by clinical evaluation.

A proportion of patients with VL in pregnancy were discharged with clinical cure, but without parasitological confirmation of cure (which is the protocol for patients with VL in pregnancy). A ToC may not have been possible because of spleen regression, or because of advanced pregnancy as contraindication for splenic aspiration with no lymphadenopathy as alternative aspirate site.

A recent report of treatment outcomes for VL from Ethiopia showed initial cure rates of 96.7% for non HIV-co-infected patients treated with AmBisome (total dose 24-35mg/kg) [22,23]. In our cohort, initial cure was achieved for only 61% of pregnant women treated with the standard dose of AmBisome (30mg/kg), while 35.6% required extension of treatment to achieve initial cure, indicating an impaired treatment response in pregnant women. This is similar to the initial cure rates of 60% for HIV co-infected patients after treatment with 30mg/kg AmBisome[24], and likely due to the modulated immune system during pregnancy. This justifies the routine ToC after the standard dose of AmBisome (of 30mg/kg) for pregnant women with VL to establish proper parasitological cure before discharge.

Many pregnant and postpartum women were discharged anaemic, despite routine treatment with ferrous sulphate and folic acid for all patients. This is a risk especially for pregnant women who may deliver at home after discharge from VL treatment.

Obstetric and perinatal outcomes

Twenty-one percent (17/81) of pregnant patients delivered during or directly after finishing treatment. Most adverse pregnancy outcomes involved miscarriages in the first trimester or premature deliveries in the third trimester. Although a proportion of the adverse pregnancy outcomes was probably caused by VL, no community-based studies are available to compare these pregnancy outcomes with the general population. It is interesting that only two adverse pregnancy outcomes (6.1%) were recorded for women with VL admitted in the second trimester. The relation between adverse pregnancy outcomes and different immunological states of the pregnant women in different trimesters remains a point for future research, as well as the relation between (different trimesters in) pregnancy and susceptibility for developing VL.

Significantly more adverse pregnancy outcomes were found among primigravidae (OR 4.33, 95% CI 1.03–18.2). A similar outcome is known for malaria infection during pregnancy, with placental malaria with adverse outcomes being more common among primigravidae, although the mechanism may be different. Much about the immunological response and pregnancy outcomes per trimester remains unknown. It would be relevant to consider studying placenta's of VL patients after delivery for the presence of placental invasion with parasites, including those who experience premature deliveries during treatment or after apparently successful treatment.

When interpreting the perinatal outcomes, it should be noted that exact gestation of pregnancy was often not known, but estimated based on history of last menstruation and assessment of the abdomen on admission and/or assessment of the baby postpartum. No documentation from early-pregnancy antenatal care visits or ultrasound reports were available. A differentiation between intra-uterine growth restriction or low birth weight due to prematurity is therefore difficult.

Data about (home) deliveries after discharge from VL treatment were not available. Because no follow-up data were available, it is possible that the actual number of adverse pregnancy

outcomes is higher than the presented figures. Moreover, no follow-up data were available for neonatal outcomes. Data about the condition of the neonate after delivery until discharge were also not available, as the neonates often had their own patient records which we were not able to identify from the archive. Preterm birth is the most important determinant of adverse infant outcomes in terms of survival and quality of life globally, and adverse outcomes are significant especially in low resource settings[25,26].

Data about birth weight and gestation at time of delivery were available for the babies of only 13 patients (out of 43 patients who delivered before or during treatment), therefore no conclusions can be drawn from these numbers. The available data show low birth weight deliveries, which is in line with the high number of premature deliveries.

Data about blood loss during delivery were not available from patients admitted postpartum after home delivery. Post-partum haemorrhage was reported in 35% of in-hospital deliveries, but did not lead to any maternal mortality, probably thanks to the presence of skilled health personnel, and availability of uterotonic drugs and blood transfusion.

Surgical evacuation for retained products of conception (either manual vacuum aspiration or curettage) was reported in 29% of cases. Retained placenta is known to occur more frequently in women with preterm vaginal delivery than in women with term vaginal delivery (9.1% vs 1.1)[27]. In VL patients it may be difficult to distinguish between fever due to VL and infected retained products of conception. In already anaemic patients, prone for infectious complications and bleeding, the threshold for clinicians to perform evacuation may have been lower than in non-VL patients. A high rate of retained products of conception in VL patients has not been reported elsewhere.

Strengths and limitations

We described a large cohort of patients diagnosed with VL in pregnancy and the early postpartum period. Previous publications about VL in pregnancy are limited in number and have smaller patient cohorts. Availability of patient records in general was good and data about patient characteristics on admission and treatment outcomes were complete in most cases.

The major limitation of the study is missing data for some of the outcomes, especially for pregnancy and neonatal outcomes, due to the retrospective study design. Pregnancy outcomes after discharge from VL treatment and neonatal outcomes after birth remain an important point for future research, ideally with prospective studies. The same accounts for final VL outcomes, as the incidence of relapse VL after initial successful treatment of VL in pregnancy is not known. However, longitudinal follow-up is very challenging in this setting in South Sudan, where patients return to their (often remote) villages after completing treatment, with important access barriers due to poor infrastructure and public transport, especially during the rainy season, and general insecurity.

This study demonstrates that in order to achieve favorable treatment outcomes a certain level of care is needed for the treatment of VL in pregnancy, with the ability to manage potential complications of the disease. We mentioned the need for treatment with AmBisome (which requires cold chain and intravenous administration), intravenous antibiotics, blood transfusion, management of obstetric complications and care for premature neonates. These services are available in the MSF health facility in Lankien in the current set-up, but may be very challenging to realize in all VL treatment sites in South Sudan, as they are typically in remote and extremely resource-limited locations with severe capacity and logistic constraints.

Conclusion

This is the largest cohort reporting patient characteristics and maternal and pregnancy outcomes for patients with VL in pregnancy treated with AmBisome. We reported high survival rates and cure rates at discharge for women with VL in pregnancy, treated in a low resource setting. However, patients with VL in pregnancy treated with AmBisome often required extension of treatment to achieve cure after the initial total standard dose of 30mg/kg. Careful evaluation of patients is needed at the end of the standard treatment regimen, in order to ensure proper (parasitological) cure before discharge.

Follow up data would be relevant to determine if women with VL in pregnancy have an increased risk for relapse of VL disease, and to understand the risk of pregnancy-related complications after discharge. Commonly reported complications, such as suspected secondary infections, severe anaemia and obstetric complications, emphasize the need for appropriate medical care in addition to VL treatment for this patient group.

Supporting information

S1 Checklist. STROBE checklist. (DOC)

Acknowledgments

We would like to thank the MSF field teams that have contributed to care of the VL patients in Lankien hospital over the past years, the MSF teams in Juba (South Sudan) and Lokichoggio (Kenya) for their support in the data collection, and Despina Pampaka for reviewing the manuscript.

Author Contributions

Conceptualization: M. Ruby Siddiqui, Koert Ritmeijer.

Data curation: Judith E. Pekelharing.

Formal analysis: Judith E. Pekelharing.

Investigation: Judith E. Pekelharing, Francis Gatluak, Tim Harrison.

Methodology: Judith E. Pekelharing, Koert Ritmeijer.

Project administration: Koert Ritmeijer.

Supervision: Fernando Maldonado, Koert Ritmeijer.

Writing – original draft: Judith E. Pekelharing.

Writing – review & editing: Tim Harrison, Fernando Maldonado, M. Ruby Siddiqui, Koert Ritmeijer.

References

- 1. Zijlstra EE, El Hassan AM. Leishmaniasis in Sudan. 3. Visceral leishmaniasis. Trans R Soc Trop Med Hyg. 2001; 95: S27–S58. https://doi.org/10.1016/s0035-9203(01)90218-4 PMID: 11370250
- WHO Expert Committee on the Control of the Leishmaniases & World Health Organization. Control of the leishmaniases: report of a meeting of the WHO Expert Committee on the Control of Leishmaniases. Geneva; 2010. Available: http://www.who.int/iris/handle/10665/44412
- 3. World Health Organization (WHO). South Sudan—Leishmaniasis country profile. 2015. Available: http://www.who.int/leishmaniasis/burden/South-Sudan_2015-hl.pdf?ua=1
- Al-salem W, Herricks JR, Hotez PJ. A review of visceral leishmaniasis during the conflict in South Sudan and the consequences for East African countries. Parasit Vectors. 2016; 1–11. https://doi.org/ 10.1186/s13071-015-1291-6
- Burza S, Croft SL, Boelaert M. Seminar Leishmaniasis. Lancet. 2018; 392: 951–970. https://doi.org/10. 1016/S0140-6736(18)31204-2 PMID: 30126638

- Melaku Y, Collin SM, Keus K, Gatluak F, Ritmeijer K, Davidson RN. Treatment of Kala-Azar in Southern Sudan using a 17-Day Regimen of Sodium Stibogluconate Combined with Paromomycin: A Retrospective Comparison with 30-Day Sodium Stibogluconate Monotherapy. Am J Trop Med Hyg. 2007; 77: 89– 94. PMID: 17620635
- Kamink SS, Collin SM, Harrison T, Gatluak F, Mullahzada W, Ritmeijer K. A clinical severity scoring system for visceral leishmaniasis in immunocompetent patients in South Sudan. PLoS Negl Trop Dis. 2017; 86: 1–15. https://doi.org/10.1371/journal.pntd.0005921
- Ruiz-postigo A, Pita J, Lado M, Ben-ismail R, Argaw D, Alvar J. Visceral Leishmaniasis Outbreak in South Sudan 2009–2012: Epidemiological Assessment and Impact of a Multisectoral Response. PLoS Negl Trop Dis. 2014; 8: e2770. https://doi.org/10.1371/journal.pntd.0002770
- 9. Mor G, Cardenas I. The Immune System in Pregnancy: A Unique Complexity. Am J Reprod Immunol. 2010; 63: 425–433. https://doi.org/10.1111/j.1600-0897.2010.00836.x PMID: 20367629
- Berger BA, Bartlett AH, Saravia NG, Galindo Sevilla N. Pathophysiology of Leishmania Infection during Pregnancy. Trends Parasitol. 2017; 33: 935–946. https://doi.org/10.1016/j.pt.2017.08.012 PMID: 28988681
- 11. Pagliano P, Carannante N, Rossi M, Gramiccia M, Gradoni L, Faella FS, et al. Visceral leishmaniasis in pregnancy: a case series and a systematic review of the literature. J Antimicrob Chemother. 2004; 1–5. https://doi.org/10.1093/jac/dkh538 PMID: 15649998
- Figueiro-Filho EA, Duarte G, El Beitune P, Quintana SM, Maia TL, Visceral leishmaniasis (kala-azar) and pregnancy. Infect Dis Obs Gynaecol. 2004; 12: 31–40. https://doi.org/10.1080/ 1064744042000210384 PMID: 15460194
- Mueller M, Balasegaram M, Koummuki Y, Ritmeijer K, Santana MR, Davidson R. A comparison of liposomal amphotericin B with sodium stibogluconate for the treatment of visceral leishmaniasis in pregnancy in Sudan. J Antimicrob Chemother. 2006; 58: 811–815. https://doi.org/10.1093/jac/dkl342 PMID: 16916865
- Miah M, Ayaz F, Maniruzzaman M, Ahasan M, Bari S, Mawla S, et al. Kala azar in Pregnancy. Mymensingh Med J. 2010; 19: 529–32. PMID: 20956894
- Adam GK, Omar SM, Ahmed MA, Abdallah TM, Ali AA. Cross--sectional study of the case-fatality rate among patients with visceral leishmaniasis infections during pregnancy in Sudan. Int J Gynaecol Obstet. 2017; 119–120. https://doi.org/10.1002/ijgo.12332 PMID: 28960291
- 16. Ministry of Health South Sudan. Guidelines for diagnosis, treatment and prevention of visceral leishmaniasis in South Sudan.
- World Health Organization (WHO). South Sudan Tuberculosis profile, 2017. Available: <u>https://extranet.</u> who.int/sree/Reports?op=Replet&name=/WHO_HQ_Reports/G2/PROD/EXT/ TBCountryProfile&ISO2=SSD&outtype=html
- Li X, Zhou X. Co-infection of tuberculosis and parasitic diseases in humans: a systematic review. Parasit Vectors. 2013; 6.
- 19. World Health Organization (WHO). Global Health Observatory data repository. 2018. Available: http:// apps.who.int/gho/data/view.main.SDG33v?lang=en
- Bogaart E Van Den, Berkhout MMZ, Nour ABYM, Mens PF, Talha AA, Adams ER, et al. Concomitant malaria among visceral leishmaniasis in-patients from Gedarif and Sennar States, Sudan: a retrospective case-control study. BMC Public Health. 2013; 13: 1–13. https://doi.org/10.1186/1471-2458-13-1
- Adam GK, Abdulla MA, Ahmed AA, Adam I. Maternal and perinatal outcomes of visceral leishmaniasis (kala-azar) treated with sodium stibogluconate in eastern Sudan. Int J Gynecol Obstet. 2009; 107: 208– 210. https://doi.org/10.1016/j.ijgo.2009.08.002 PMID: 19766208
- Tamiru A, Tigabu B, Yifru S, Diro E, Hailu A. Safety and efficacy of liposomal amphotericin B for treatment of complicated visceral leishmaniasis in patients without HIV, North-West Ethiopia. BMC Infect Dis. 2016; 16: 548. https://doi.org/10.1186/s12879-016-1746-1 PMID: 27724891
- Gebreyohannes EA, Bhagvathula AS, Abegaz TM, Seid MA. Treatment outcomes of visceral leishmaniasis in Ethiopia from 2001 to 2017: a systematic review and meta-analysis. Infect Dis Poverty. 2018; 7: 1–9. https://doi.org/10.1186/s40249-017-0384-1
- Ritmeijer K, Horst R, Chane S, Aderie EM, Piening T, Collin SM, et al. Limited Effectiveness of High-Dose Liposomal Amphotericin B (AmBisome) for Treatment of Visceral Leishmaniasis in an Ethiopian Population With High HIV Prevalence. Clin Infect Dis. 2011; 53: e152–8. https://doi.org/10.1093/cid/ cir674 PMID: 22016502
- World Health Organization (WHO). WHO recommendations on interventions to improve preterm birth outcomes. 2015. PMID: 26447264

- 26. Dornemann J, van den Boogaard W, Van den Bergh R, Takarinda KC, Martinez P, Bekouanebandi JG, et al. Where technology does not go: specialised neonatal care in resource-poor and conflict-affected contexts. Public Heal Action. 2017; 7: 168–174. https://doi.org/10.5588/pha.16.0127 PMID: 28695092
- Romero R, Chiung YH, Athanassiadis AP, Hagay Z, Avila C, Nores J, et al. Preterm delivery: A risk factor for retained placenta. Am J Obstet Gynaecol. 1990; 163: 823–825.